# TECHNICAL NOTE

# State of Pine Decline in the Southeastern United States

# Lori Eckhardt, Mary Anne Sword Sayer, and Don Imm

Pine decline is an emerging forest health issue in the southeastern United States. Observations suggest pine decline is caused by environmental stress arising from competition, weather, insects and fungi, anthropogenic disturbances, and previous management. The problem is most severe for loblolly pine on sites that historically supported longleaf pine, are highly eroded, or are not managed. The purposes of this technical note are (1) to describe the symptomology and extent of pine decline in the southeastern United States; (2) to describe its connection with root disease, resource stress, and silviculture; and (3) to summarize the consensus opinion of scientists and land managers during a workshop sponsored by the US Army Strategic Environmental Research and Development Program regarding the scope of this syndrome and the best research avenues to counter its potential effect on the sustainability of southern pine forests.

Keywords: Leptographium, loblolly die-off, off-site, root disease, resource stress

# **Observations of PD**

veral reports have suggested that localized forest health problems are increasing in southern pine forests (Otrosina et al. 1999, Hess et al. 2005, Eckhardt et al. 2007). When these events are accompanied by sparse and chlorotic crowns, low annual stemwood production, and isolation of fungal pathogens from roots other than Phytophthora cinnamomi Rands or Heterobasidion annosum (Fr.) Bref., they are commonly referred to as pine decline (PD) (Otrosina et al. 1999, Eckhardt et al. 2007). Mortality may occur within 3 years after symptoms (Hess et al. 2002, Eckhardt et al. 2007, Menard 2007). PD has been observed across the southeastern United States from Alabama to South Carolina in the Atlantic and East Gulf Coastal Plains and Piedmont Province, as well as the fall-line Sandhills interfacing these regions. It has primarily been reported on public lands managed for multiple objectives that include but do not emphasize timber production (Hess et al. 2002, Menard 2007). PD has also been observed in southern pine forests managed by nonindustrial private landowners and forest industries (Eckhardt et al. 2007). The majority of these events have occurred in mature loblolly pine (Pinus taeda L.) and mixtures of mature loblolly and shortleaf pine (P. echinata Mill.). Loblolly pine is considered "off-site" at many of these locations that historically supported longleaf pine (P. palustris Mill.) (Hess et al. 1999, Hess et al. 2002). These observations have occurred with a decrease in the growth rate of southern pine forests over the last decade (Gadbury et al. 2004).

## The PD Setting

Most of the Atlantic and East Gulf Coastal Plains, lower Piedmont Province, and fall-line Sandhills have similar land-use histories (US Forest Service 1988, Barrett 1995). Prior to European settlement, these areas were periodically burned by Native Americans or were ignited naturally by lightning (Frost 2006). Vegetation was fire-tolerant, consisting of woodlands and sparsely treed savannas with a mixed pine–oak canopy. During the 18th and 19th centuries, European settlers cleared most of the arable land for subsistence agriculture. Erosion and nutrient depletion quickly reduced crop yields and agriculture was abandoned. Afterward, some areas were naturally seeded with loblolly pine and other early successional forest types, but most of the landscape became severely eroded by lack of vegetation. During the early and mid-20th century, federal and state agencies rehabilitated many eroded areas by planting loblolly pine (US Forest Service 1988). To their credit, much of the erosion ceased, but some of these forests remained unproductive (Ward and Mistretta 2002, Gadbury et al. 2004).

Reports of mortality attributed to PD in the early 1970s recognized that its occurrence was tied to unusual physiological and environmental conditions (Roth and Peacher 1971, Miller 1979). Sparse, chlorotic crowns and poor diameter growth, for example, were accompanied by exceptionally large cone crops produced 1 year before mortality and high amounts of lateral root deterioration and fine root mortality before crown symptoms appeared (Brown and McDowell 1968, Roth and Peacher 1971, Miller 1979). It was concluded that root pathogens and adverse soil-water relationships were contributing factors (Brown and McDowell 1968).

Between 1953 and 1999, the average annual mortality of southern pine more than doubled (Conner and Hartsell 2002). It was suggested that this trend is attributed to the lower rate of mortality expected as young stands established in the late 1960s and early 1970s developed, and subsequently, the higher rate of mortality expected when portions of these stands were not actively managed.

Copyright © 2010 by the Society of American Foresters.

Manuscript received August 24, 2009; accepted September 3, 2009.

Lori Eckhardt (eckhalg@auburn.edu), School of Forestry and Wildlife Sciences, Auburn University, 602 Duncan Drive, Suite 3301, Auburn, AL 36849-5418. Mary Anne Sword Sayer, US Forest Service, Southern Research Station, Pineville, LA 71360. Don Imm, US Fish and Wildlife Service, Panama City, FL 32405. The authors thank the US Army Strategic Environmental Research and Development Program Office and the Ecological Society of America for initiating and organizing this dialogue among scientists and land managers in the South. Appreciation is also extended to several individuals for their participation in this discussion (Lindsay Boring, Joseph W. Jones Ecological Research Center; Tom Darden, US Forest Service; Steve Davis, US Army; Todd Engstrom, University of Florida; Victor Ford, University of Arkansas; Bob Larimore, US Army; Ken McLeod, Savannah River Ecology Laboratory; Susan Ustin, University of California).

Correlation between mortality and an absence of forest management is supported by the fact that 92% of this mortality occurred in naturally regenerated stands, whereas only 8% of it was in planted stands. Overlap between the timing of PD observations and both lower growth rates and accelerated mortality, as well as the tendency for PD to occur in off-site or disturbed settings, or by age 50 years in unthinned stands (Hess et al. 1990, Otrosina et al. 1999, Menard 2007), suggests that one or more resource limitations contribute to this problem.

Recent descriptions of the disease complex associated with PD on public and private land in central Alabama (Eckhardt et al. 2007) and at Fort Benning Military Reserve in central Georgia (Menard 2007) suggest that this problem should be monitored to ensure that the sustainability of southern pine forests is not at risk. Caution is warranted because at broader scales, PD would have serious landscape implications and an effect on local economies that depend on the forest products industry. Based on current pathological examination of forests experiencing PD, this problem appears to be caused by (1) the combined effect of multiple stressors rising from site quality, climate variation, and past and current land use; (2) opportunistic insect damage and root pathogens that impair tree responses to these stressors; (3) a higher susceptibility of mature or maturing stands; and (4) in some situations, imbalances in normal populations of insect pests and root-infesting fungi.

#### Role of Root Disease in PD

Often, increases in both insect pests and the isolation of fungal pathogens from tree roots coincide with PD symptoms. Three root pathogens have been implicated as factors contributing to PD: P. cinnamomi, H. annosum, and Leptographium spp. Although the symptoms of littleleaf disease, caused by P. cinnamomi, are similar to those of PD, several attempts to link P. cinnamomi to PD resulted in only a weak relationship (Roth and Peacher 1971, Miller 1979, Mistretta and Starkey 1982). One exception was reported by Hess et al. (1999) in their reassessment of the causal factors of PD in the Talladega National Forest (TNF) in Alabama. They concluded that PD was caused by a combination of edaphic disturbances and root disease attributed to P. cinnamomi and Pythium spp. Although Leptographium spp. were isolated from roots at this location, their pathogenicity was considered secondary to that of P. cinnamomi and Pythium spp. When evaluations of PD on the TNF were repeated in 2000 through 2002, Leptographium spp. rather than P. cinnamomi and Pythium spp. were the primary root pathogens associated with PD (Hess et al. 2005). Accounts of H. annosum occurring with PD are either absent or negligible (Roth and Peacher 1971, Miller 1979, Hess et al. 2005). One exception to this generalization was reported in the early 1980s by Mistretta and Starkey (1982) who surveyed pine mortality on the Shoal Creek Ranger District of the TNF, and the Bankhead and Black Warrior Ranger Districts of the Bankhead National Forest (BNF) in Alabama. They found H. annosum sporophores on several stumps and dead loblolly and slash pine trees in the two BNF ranger districts but did not find signs of this pathogen on the TNF ranger district. Elsewhere, Otrosina et al. (1999) found both Leptographium spp. and H. annosum in declining 35-year-old longleaf pine stands in South Carolina but suggested this did not necessarily indicate that the trees were experiencing annosum root disease.

*Leptographium* spp. and their root-feeding beetle vectors are a consistent component of PD (Brown and McDowell 1968, Otrosina et al. 1999, Eckhardt et al. 2007). Root-feeding beetles are

attracted by secondary metabolites (Hodges et al. 1979). In addition to root-feeding injury, there is evidence that the pathogenicity of Leptographium spp. increases as beetles begin to feed because of changes in the porportions of oleoresin compounds produced by the tree (Eckhardt et al., unpublished). This new oleoresin environment favors the growth of the fungus (Paine et al. 1997). Successful colonization by Leptographium spp., in turn, appears to benefit the development of insect brood within the trees' roots because the fungus either serves as a food source for emerging larvae or makes the environment more habitable (Eckhardt et al. 2004b). The insect-fungus relationship expands the root disease, and the insects are indispensable in vectoring fungal spores from infected roots to healthy roots. In some situations, however, the fungus alone may be a major contributor to PD because it obstructs root vascular tissue (Eckhardt et al. 2004a), which reduces the translocation of water and mineral nutrients within the root system.

#### **Role of Resource Stress in PD**

In addition to *Leptographium* root disease, resource limitations arising from the disruption of natural stress avoidance mechanisms may trigger or accelerate PD. A downward shift in whole-crown leaf area, for example, sustains foliar physiology when soil fertility or water availability is suboptimal (Vose and Allen 1988, Pallardy et al. 1995). The elasticity of leaf area ensures that nutrition and water are adequate at the cellular level. Trees may endure resource stress as long as leaf area maintains a neutral or positive whole-tree carbon balance, thus producing enough energy to sustain cellular processes. Vigor may be jeopardized, however, by a drop in leaf area below a critical level or persistent resource deficiencies at the cellular level.

Resource stress may also be evaded or reduced by establishing a pattern of carbon allocation that favors root system growth. This has been demonstrated for loblolly pine on sandy and loamy soils in North Carolina (Hacke et al. 2000) and for longleaf pine on xeric and mesic sites in Georgia (Addington et al. 2006). At both locations, the ratio of surface areas of absorbing roots and foliage was greater on sites with lower plant-available water. This pattern of carbon allocation contributed to the maintenance of stomatal conductances on drier sites that were comparable to those on more moist sites. Water uptake may also rely on deep roots capable of supplying water for immediate aboveground processes or for hydraulic redistribution (Hacke et al. 2000, Domec et al. 2004, Warren et al. 2007). Without hydraulic redistribution, very dry soils may cause root dehydration and the localized cavitation of root xylem (Domec et al. 2004). After drought relief, some lost xylem function could be restored (Domec et al. 2004), but it may not be completely regained until new roots are grown (Hacke et al. 2000).

Together with broad windows of leaf area and root system adjustment, the internal recycling of mineral nutrients also sustains cellular processes when their availability is low (Nambiar and Fife 1991, Oren and Sheriff 1995). In addition to resorption of nutrients before the senescence of older foliage (Dalla-Tea and Jokela 1994), the potential exists for nutrient accumulation in and translocation from nonsenescent foliage, followed by replenishment (Nambiar and Fife 1991, Marschner 1995). The mobility of mineral nutrients with a role in stomatal function and osmotic adjustment, for example, may be key to maintaining photosynthate production and transport and dehydration tolerance during drought (Kramer and Boyer 1995, Cakmak 2005).

## Role of Silviculture in PD

Silvicultural choices have the potential to interact with a tree's response to resource stress. If a stand's carrying capacity is exceeded, for example, thinning by natural or prescription-based means increases the likelihood that essential resources will be adequate for stress avoidance. Although density-related mortality is expected at around 50% of maximum stand density index for loblolly pine (Dean and Baldwin 1993), subdominant trees may persist beyond this point. For example, the relative stand density index of overstocked loblolly pine in Louisiana was approximately 85% between age 11 and 13 years, and self-thinning did not proceed until significant water deficit was reached at age 14 years (Sword Sayer et al. 2004).

Once a stand density threshold is reached and if self-thinning is delayed, canopy properties that control whole-crown carbon fixation become insensitive to stand density (Dean and Baldwin 1996, Dean 2001). Similarly, when stands are overstocked, the relationship between canopy properties and root system growth is poor, and carbon allocation to fine roots may be insufficient for adequate soil resource uptake (Dean 2001). These observations suggest that without timely moderation of a stand's carrying capacity, the vigor of residual trees will deteriorate and compromise inherent stress avoidance mechanisms that depend on root system function.

In some situations, the potential also exists for repeated fire to indirectly interfere with root system function and therefore a stand's natural mechanisms of resource stress avoidance. The survival and production of healthy roots is dependent on the supply of recently fixed carbon (van den Driessche 1987, Dickson 1991). In turn, the root system serves as a reservoir of stored starch (Gholz and Cropper 1991, Sword et al. 2000) that is mobilized to support developing foliage (Dickson 1991). The negative effect of defoliation on stemwood production appears to be season-dependent (Johansen and Wade 1987, Weise et al. 1987). Likewise, crown scorch may have a variable effect on carbon sinks other than stemwood growth depending on the time of years when fire occurs (Sword Sayer et al. 2006). In Louisiana, for example, severe scorch in September reduced the root starch content of longleaf pine for 17 months (Sword and Haywood 1999). In Georgia, Guo et al. (2004) found that severe scorch in June also reduced the root starch content of longleaf pine but did not affect this species' fine root biomass. Under normal circumstances, a decrease in root starch may have negligible effects on long-term root production and tree vigor. However, if the re-establishment of foliage after scorch is constrained by the availability of root starch, subsequent foliage and root responses could hinder one or more stress avoidance mechanisms.

# Workshop Consensus and Recommendations

A 3-day workshop in June of 2007 was sponsored by the US Army Strategic Environmental Research and Development Program Office. The overall goal of this workshop was to assess the extent and cause of localized declines in southern pine health and identify short- and long-term remedial management actions. Scientists and land managers participated in four discussions of the problem that led to the preparation of three reports and one nontechnical summary (Ecological Society of America 2008). The purpose of one of these discussions was to come to consensus regarding the scope of PD and the research needed to counter its potential effect on southern pine forest health. Before these points could be addressed, workshop participants indicated that two types of information were necessary. First, it is not known whether PD is new, cyclic, or related to climate change. Second, the locales, species, and forest types exhibiting PD are not well documented. We must first understand whether the problem is new, cyclic, or climate-related before methods of detection, risk analysis, and mitigation can be developed. Once these tools are developed, the locales, species, and forest types susceptible to PD can be defined. With this information, PD can be predicted and mitigation can be prioritized to favor those situations that are the most recoverable. Two caveats are attached to the success of this plan. First, the effect of climate on PD may make it difficult to isolate causal symptomology. Second, management opportunities to sustain forest health may be limited by environmental policies and land-use demands.

Group deliberations identified three activities that would be helpful as PD research is designed:

- Current forest health monitoring (FHM) variables, frequencies, and intensities should be evaluated to determine whether they are adequate to quantify the scope, severity, and spread of PD, and these efforts should be integrated among established regional Forest Inventory Analysis, FHM, and Eastern Forest Threat programs directed by the US Forest Service.
- Management effects on critical resources linked to PD should be evaluated, and management practices showing promise as tools to reduce stressors linked to PD should be investigated.
- Forest conditions that could worsen the regional effect of PD, such as the presence of invasive exotic pests, should be investigated.

## **Recommended Actions**

Based on workshop consensus, several lines of research are needed to understand and respond to PD in the southeastern United States:

- 1. A current and extensive geographic record of locales exhibiting forest health problems related and unrelated to PD but experiencing mortality should be maintained at the regional scale and made available for the development of models that predict the occurrence, spread, and probability of PD.
- 2. Models that predict spatial and temporal patterns of speciesand forest system–specific pathogenic infection and spread and tree mortality rates should be developed.
- 3. A better understanding of southern pine physiological responses to interacting stressors associated with PD is needed.
- Scale- and forest setting–appropriate remote sensing technology that provides information regarding forest health and tree mortality is needed.

## Summary

At present, PD is a localized forest health issue caused by environmental stress arising from interaction among several abiotic and biotic factors. There is not enough information about PD at this point to recognize it as a regional forest health threat. However, because PD has been observed among several southern pine species and at locations across the southeastern United States, the scope of this problem has the potential to be broad. We need to know whether PD is new, cyclic, or related to climate change, and we need to know how it correlates to certain locales, species, and forest types to determine its scope. The magnitude of PD may be tempered or exaggerated by changing regional forces (e.g., climate, land-use, invasive species). Therefore, evaluation of this problem requires simultaneous investigation of stand environment and management history. Because the ecological and economic consequences of PD have the potential to adversely affect the sustainability of southern pine forests, research is urgently needed in several areas. Immediate steps should first assess forest health and mortality rates across the southern pine region and develop models that predict the extent and probability of PD. Second, information about the environmental and physiological conditions and management practices that predispose stands to PD is needed. With this, remote-sensing tools that communicate a stand's susceptibility to PD and the expected rate of decline-induced mortality can be created.

## Literature Cited

- ADDINGTON, R.N., L.A. DONOVAN, R.J. MITCHELL, J.M. VOSE, S.D. PECOT, S.B. JACK, U.G. HACKE, J.S. SPERRY, AND R. OREN. 2006. Adjustments in hydraulic architecture of *Pinus palustris* maintain similar stomatal conductance in xeric and mesic habitats. *Plant Cell Environ*. 29:535–545.
- BARRETT, J.W. 1995. *Regional silviculture of the United States*, 3rd ed. John Wiley and Sons, New York. 643 p.
- BROWN, H.D., AND W.E. MCDOWELL. 1968. Status of loblolly pine die-off on the Oakmulgee District, Talladega National Forest, Alabama—1968. US For. Serv., Southern Region, State and Private Forestry Rep. No. 69-2-28. 7 p.
- CAKMAK, I. 2005. The role of potassium in alleviating detrimental effects of abiotic stresses in plants. J. Plant Nutr. 168:521–530.
- CONNER, R.C., AND A.J. HARTSELL. 2002. Forest area and conditions. P. 357–401 in *Southern forest resource assessment,* Wear, D.N., and J.G. Gries (eds.). US For. Serv., South. Res. Stn., Asheville, NC.
- DALLA-TEA, F., AND E.J. JOKELA. 1994. Needlefall returns and resorption rates of nutrients in young intensively managed slash and loblolly pine stands. *For. Sci.* 40:650–662.
- DEAN, T.J. 2001. Potential effect of stand structure on belowground allocation. For. Sci. 47:69–76.
- DEAN, T.J., AND V.C. BALDWIN, JR. 1993. Using a density-management diagram to develop thinning schedules for loblolly pine plantations. US For. Serv. Res. SO-275. 7 p.
- DEAN, T.J., AND V.C. BALDWIN. 1996. Growth in loblolly pine plantations as a function of stand density and canopy properties. *For. Ecol. Manag.* 82:49–58.
- DICKSON, R.E. 1991. Assimilate distribution and storage. P. 51–85 in *Physiology of trees*, Raghavendra, A.S., (ed.). John Wiley & Sons, New York. 509 p.
- DOMEC, J.-C., J.M. WARREN, F.C. MEINZER, J.R. BROOKS, AND R.E. COULOMBE. 2004. Native root xylem embolism and stomatal closure in stands of Douglas-fir and ponderosa pine: Mitigation by hydraulic redistribution. *Oecologia* 141:7–16.
- ECKHARDT, L.G., J.P. JONES, AND K.D. KLEPZIG. 2004a. Pathogenicity of *Leptographium* species associated with loblolly pine decline. *Plant Dis.* 88:1174-1178.
- ECKHARDT, L.G., R.A. GOYER, K.D. KLEPZIG, AND J.P. JONES. 2004b. Interactions of *Hylastes* species (Coleoptera: Scolytidae) with *Leptographium* species associated with loblolly pine decline. *J. Econ. Entomol.* 97:468–474.
- ECKHARDT, L.G. AND R.D. MENARD. (unpublished). Oleoresin compound levels in relation to infection by bluestain fungi vectored by root-feeding bark beetles.
- ECKHARDT, L.G., A.M. WEBER, R.D. MENARD, J.P. JONES, AND N.J. HESS. 2007. Insect–fungal complex associated with loblolly pine decline in central Alabama. *For. Sci.* 53:84–92.
- ECOLOGICAL SOCIETY OF AMERICA 2008. *ESA Forest Health Workshop*. Available online at www.esa.org/science\_resources/forest\_health\_workshop.php; last accessed Jan. 22, 2010.
- FROST, C.C. 2006. History and future of the longleaf pine. P. 9–42 in *The Longleaf pine ecosystem, ecology, silviculture, and restoration,* S. Jose, E.J. Jokela, and D.L. Miller (eds). Springer Science+Business Media, Inc., New York. 438 p.
- GADBURY, G.L., M.S. WILLIAMS, AND H.T. SCHREUDER. 2004. Revisiting the southern pine growth decline: Where are we 10 years later? US For. Serv. Gen. Tech. Rep. RMRS-GTR-124. 10 p.
- GHOLZ, H.L., AND W.P. CROPPER. 1991. Carbohydrate dynamics in mature *Pinus* elliottii var. elliottii trees. Can. J. Forest Res. 21:1742–1747.
- GUO, D.L., R.J. MITCHELL, AND J.J. HENDRICKS. 2004. Fine roots branch orders respond differently to carbon source-sink manipulations in a longleaf pine forest. *Oecologia* 140:450–457.
- HACKE, U.G., J.S. SPERRY, B.E. EWERS, D.S. ELLSWORTH, K.V.R. SCHAFER, AND R. OREN. 2000. Influence of soil porosity on water use in *Pinus taeda*. *Oecologia* 124:495–505.
- HESS, N.J., A. MANGINI, D. STARKEY, AND R.C. KERTZ. 1990. Evaluation of mortality in unthinned loblolly plantations on the Bankhead Ranger District. US For. Serv., Southern Region, State and Private Forestry Rep. No. 90-2-13. 16 p.

- HESS, N.J., W.J. OTROSINA, J.P. JONES, A.J. GODDARD, AND C.H. WALKINSHAW. 1999. Reassessment of loblolly pine decline on the Oakmulgee District, Talladega National Forest, Alabama. P. 560–564 in *Proc. of Tenth Biennial Southern Silvicultural Research Conference*, Haywood, J.D. (ed.). US For. Serv., South. Res. Stn., Ashville, NC.
- HESS, N.J., W.J. OTROSINA, E.A. CARTER, J.R. STEINMAN, J.P. JONES, L.G. ECKHARDT, A.M. WEBER, AND C.H. WALKINSHAW. 2002. Assessment of loblolly pine decline in central Alabama. P. 558–564 in *Proc. of Eleventh Biennial Southern Silvicultural Research Conference*, Outcalt, K.W. (ed.). US For. Serv., South. Res. Stn., Asheville, NC.
- HESS, N.J., L.G. ECKHARDT, R.D. MENARD, A.J. GODDARD, AND E.A. CARTER. 2005. Assessment of loblolly pine decline on the Shoal Creek/Talladega Ranger Districts, Talladega National Forest, Alabama and Choccolocca State Forest. US For. Serv., FHP Rep. No. 2005-02-05. 36 p.
- HODGES, J.D., W.W. ELAM, W.F. WATSON, AND T.E. NEBERKER. 1979. Oleoresin characteristics and susceptibility of four southern pines to southern pine beetle (Coleoptera: Scolytidae) attacks. *Can. Entomol.* 111:889–896.
- JOHANSEN, R.W., AND D.D. WADE. 1987. Effects of crown scorch on survival and diameter growth of slash pines. *South. J. For.* 11:180–184.
- KRAMER, P.J., AND J.S. BOYER. 1995. Water relations of plants and soils. Academic Press, Inc., New York. 495 p.
- MARSCHNER, H. 1995. *Mineral nutrition of higher plants*, 2nd ed. Elsevier, Ltd., Amsterdam, The Netherlands. 889 p.
- MENARD, R.D. 2007. An assessment of loblolly pine decline risk mapping system for the use of managing loblolly pine decline sites within Red Cockaded Woodpecker (RCW) habitat. MSc thesis, Louisiana State University, Baton Rouge, LA. 128 p.
- MILLER, R.E. 1979. Loblolly pine die-off status report. US For. Serv. Southeastern Area, State and Private Forestry Rep. No. 79-2-8. 3 p.
- MISTRETTA, P.J., AND D. STARKEY. 1982. Evaluation of spot dieout on three northern districts of the National forests in Alabama. US For. Serv., Southern Region, State and Private Forestry Rep. No. 83-2-3. 3 p.
- NAMBIAR, E.K., AND D.N. FIFE. 1991. Nutrient retranslocation in temperate conifers. *Tree Physiol.* 9:185–207.
- OREN, R., AND D.W. SHERIFF. 1995. Water and nutrient acquisition by roots and canopies. P. 39–74 in *Resource physiology of conifers: Acquisition, allocation, and utilization,* Smith, W.K., and T.M. Hinckley (eds.). Academic Press, Inc., New York.
- OTROSINA, W.J., D. BANNWART, AND R.W. RONCADORI. 1999. Root-infecting fungi associated with a decline of longleaf pine in the southeastern United States. *Plant Soil.* 217:145–150.
- PAINE, T.D., K.F. RAFFA, AND T.C. HARRINGTON. 1997. Interactions among scolytid bark beetles, their associated fungi, and live host conifers. *Annu. Rev. Entomol.* 42:179–206.
- PALLARDY, S.G., J. CERMAK, F.W. EWERS, M.R. KAUFMANN, W.C. PARKER, AND J.S. SPERRY. 1995. Water transport dynamics in trees and stands. P. 301–389 in *Resource physiology of conifers: Acquisition, allocation, and utilization, Smith,* W.K., and T.M. Hinckley (eds.). Academic Press, Inc., New York.
- ROTH, E.R., AND P.H. PEACHER. 1971. Alabama loblolly pine die-off evaluation. US For. Serv., Southeastern Area, State and Private Forestry Rep. No. 72-2-9. 3 p.
- SWORD, M.A., AND J.D. HAYWOOD. 1999. Effects of crown scorch on longleaf pine fine roots. P. 223–227 in Proc. of Tenth Biennial Southern Silvicultural Research Conference, Haywood, J.D. (ed.). US For. Ser., South. Res. Stn., Asheville, NC.
- SWORD, M.A., E.A. KUEHLER, AND Z. TANG. 2000. Seasonal fine roots carbohydrate relations of plantation loblolly pine after thinning. J. Sustain. Forest. 10:295–305.
- SWORD SAYER, M.A., J.C.G. GOELZ, J.L. CHAMBERS, Z. TANG, T.J. DEAN, J.D. HAYWOOD, AND D.J. LEDUC. 2004. Long-term trends in loblolly pine productivity and stand characteristics in response to thinning and fertilization in the West Gulf region. *For. Ecol. Manag.* 192:71–96.
- SWORD SAYER, M.A., J.C.G. GOELZ, AND J.D. HAYWOOD. 2006. Effects of prescribed fire on production of foliage by sapling longleaf pine. P. 223–227 in *Proc. of Thirteenth Biennial Southern Silvicultural Research Conference*, Connor, K. (ed.). US For. Serv., South. Res. Stn., Asheville, NC.
- US FOREST SERVICE. 1988. *The South's fourth forest*. Forest Resource, US For. Serv. Rep. No. 24. US For. Serv., Washington, DC. 512 p.
- VAN DEN DRIESSCHE, R. 1987. Importance of current photosynthate to new root growth in planted conifer seedlings. *Can. J. For. Res.* 17:776–782.
- VOSE, J.M., AND H.L. ALLEN. 1988. Leaf area, stemwood growth, and nutrition relationships in loblolly pine. *For. Sci.* 34:547–563.
- WARD, J.D., AND P.A. MISTRETTA. 2002. Impacts of pests on forest health. P. 403–428 in Southern forest resource assessment, Wear, D.N., and J.G. Gries (eds.). US For. Serv., South. Res. Stn., Asheville, NC.
- WARREN, J.M., F.C. MEINZER, J.R. BROOKS, J.C. DOMEC, AND R. COULOMBE. 2007. Hydraulic redistribution of soil water in two old-growth coniferous forests: Quantifying patterns and controls. *New Phytol.* 173:753–765.
- WEISE, D.R., R.W. JOHANSEN, AND D.D. WADE. 1987. Effects of spring defoliation on first-year growth of young loblolly and slash pines. US For. Serv. Res. Note SE-347. 4 p.